Author’s response to reviews

Title: Molecular tests for human papillomavirus (HPV), Chlamydia trachomatis and Neisseria gonorrhoeae in liquid-based Pap cytology specimen

Authors:

Sin Hang Lee (sinhang.lee@milfordhospital.org)
Veronica S Vigliotti (veronica.vigliotti@milfordhospital.org)
Suri Pappu (suri.pappu@milfordhospital.org)

Version: 5 Date: 20 March 2009

Author’s response to reviews:

Dear Editor:
Thank you for your letter of March 18, 2009. The manuscript is revised and shortened, now with a word count of 2981, excluding abstract, conclusions, illustrations, legends and table, etc. The revised manuscript is re-submitted for your consideration.
Sincerely,

Sin Hang Lee, MD
On behalf of all co-authors